Author's response to reviews

Title: Qualitatively and quantitatively similar effects of active and passive maternal tobacco smoke exposure on in utero mutagenesis at the HPRT locus

Authors:

Stephen G Grant (sgg@pitt.edu)

Version: 4 Date: 23 May 2005

Author's response to reviews:

This revision affects ONLY the title. "Similar" is not a very evocative description of the result, but stronger wording (such as "identical") is not justified or appropriate. Therefore I am annotating the word by referring to how the results are "similar": both qualitatively in the spectrum of mutational mechanisms, and quantitatively, in that the mutation frequencies are not significantly different.